Author's response to reviews

Title: Factors associated with anti-phenolic glycolipid-I seropositivity among the household contacts of leprosy cases

Authors:

   Ana Paula M Carvalho (anapaulamcarvalho@yahoo.com.br)
   Angélica COC Fabri (angelicafabri@yahoo.com.br)
   Rodrigo Correa-Oliveira (correa@cpqrr.fiocruz.br)
   Francisco CF Lana (xicolana@ufmg.br)

Version: 2
Date: 31 October 2014

Author's response to reviews:

Dear editor,

We did the changes requested.

We put the line and page numbers in the new version of the manuscript.

We also included in the Methods section information about informed consent for participation in the study (page 6, lines 160-163).

Sincerely,

The authors